Author's response to reviews

Title: Does physiotherapy reduce the incidence of postoperative complications in patients following pulmonary resection via thoracotomy? A protocol for a randomised controlled trial

Authors:

Julie C Reeve (julie.reeve@aut.ac.nz)
Kristine Nicol (KristineN@adhb.govt.nz)
Kathy Stiller (Kathy.Stiller@health.sa.gov.au)
Kathryn M McPherson (katmcphe@aut.ac.nz)
Linda Denehy (l.denehy@unimelb.edu.au)

Version: 2 Date: 3 July 2008

Author's response to reviews:

Endnote discrepancies rechecked as requested. Unable to determine differences between number of references in text and in reference list. In our document these correlate.

Section headings altered as requested.

Tables placed after reference list as requested. Minor typographical changes made.

Figures closely cropped as requested.